Author’s response to reviews

Title: An Intervention for Pulmonary Rehabilitators to Develop a Social Identity for Patients Attending Exercise Rehabilitation: A Feasibility and Pilot Randomised Control Trial Protocol

Authors:

Andrew Levy (levya@edgehill.ac.uk)

Bashir Matata (bashir.matata@lhch.nhs.uk)

Samantha Pilsworth (samantha.pilsworth@lhch.nhs.uk)

Adrian McGonigle (adrian.mcgonigle@lhch.nhs.uk)

Lyndsey Wigelsworth (lyndsey.wigelsworth@lhch.nhs.uk)

Linda Jones (linda.jones@lhch.nhs.uk)

Nicola Pott (nicola.pott@lhch.nhs.uk)

Max Bettany (max.bettany@lhch.nhs.uk)

Adrian Midgley (midglead@edgehill.ac.uk)

Version: 1 Date: 09 Jan 2018

Author’s response to reviews:

Dear Editors,

My co-authors and I would like to thank you for the opportunity to revise and resubmit our manuscript entitled “An Intervention for Pulmonary Rehabilitators to Develop a Social Identity for Patients Attending Exercise Rehabilitation: A Feasibility and Pilot Randomised Control Trial Protocol” (Ref: PAFS-D-00155).

The revised manuscript has been uploaded to the system. We would like to thank the reviewers for their very positive comments. To facilitate the review process, in an attached document we have included the original reviews and inserted our response after each comment. We have also highlighted, via track changes, the changes made in the revised manuscript. We have made every
effort to address all points. However, if more changes are necessary, please let us know, and we will be more than happy to make them in a second revision.

Best wishes

Andrew Levy